Author's response to reviews

Title: Congenital toxoplasmosis in a prenatal care State program

Authors:

Mariza M Avelino (mariza.avelino@gmail.com)
Ana M Castro (amaria@iptsp.ufg.br)
Waldemar N Amaral (Waldemar@sbus.org.br)
Isolina M.X Rodrigues (isolinarodrigues@uol.com.br)
Alan R Rassi (alanrassi@ipvisao.com.br)
Maria B.F Gomes (mbfranco@terra.com.br)
Tatiane L Costa (tatiane_luiza@yahoo.com.br)

Version: 3 Date: 13 February 2013

Author's response to reviews: see over
Editor-in-Chief: BMC Infectious Diseases

Dear Editor,

This study aims to evaluate whether treatment of pregnant women associated with the lack of monitoring of seroconversion for toxoplasmosis affected the prognosis of patients.

We would like to publish this article in the renowned magazine, in view of the large number of readers and statistical spread of toxoplasmosis in our state. On behalf of the group, would like to inform you that the data are not submitted to any other journal until receipt of your response. All authors participated and contributed to this study and they agree with the publish in this magazine, and there is no conflict of interest between authors.

We are attaching a copy of the manuscript entitled “CONGENITAL TOXOPLASMOSIS IN A PRENATAL CARE STATE PROGRAM” and we would like to submit to publication at your Journal as an original article. We thank your attention and look forward to your reply.

We would like to suggests the reviewers:

Marina Clare Vinaud – Laboratory studies of host parasite relationship/IPTSP/UFG - Goiânia -Brazil – mvinaud@yahoo.com.br;

M.H. Bessières – Department of parasitology CHU-Rangueil – Toulouse - France bessiere.m@chu-toulouse.fr;

Yours Faithfully,

Ph.D. Mariza Martins Avelino

Corresponding author:
Ph.D. Mariza Martins Avelino - mariza.avelino@gmail.com
telephone 5562 3241 2937, 5562 9973 1668.
Rua 235 esq com 1ª Av. s/n Setor Leste Universitário, Goiânia-GO CEP: 74605-050.